Research Paper



Impaired Upper Limb Motor Function in Post-stroke Patients and Its Impact on Trunk Control and Mobility: A Cross-sectional Study

Erum Sattar¹, Uzair Ahmad², Muheebur Rehman³, Muhammad Adnan³, Subhan Ur Rehman², Irum Shahid⁴, Abdullah³

- 1. Department of Rehabilitation Science, Ahmad Medical Institute, Peshawar, Pakistan.
- 2. Department of Physiotherapy, College of Physical Therapy, Northwest Institute of Health Sciences, Peshawar, Pakistan.
- 3. Department of Health and Biological Science, Faculty of Physical Therapy Abasyn University, Peshawar, Pakistan.

4. Rafsan Neuro Rehabilitation Center, Peshawar, Pakistan.



Citation Sattar E, Ahmad U, Rehman M, Adnan M, Rehman AU, Shahid I, Abdullah. Impaired Upper Limb Motor Function in Post-stroke Patients and Its Impact on Trunk Control and Mobility: A Cross-sectional Study. Iranian Rehabilitation Journal. 2023; 21(2):283-290. http://dx.doi.org/10.32598/irj.21.2.1834.1

doi http://dx.doi.org/10.32598/irj.21.2.1834.1



Article info: Received: 17 Jun 2022 Accepted: 06 May 2023 Available Online: 01 Jun 2023

Keywords:

Upper extremity, Motor skills, Stroke, Torso, Mobility limitation

ABSTRACT

Objectives: This study aims to investigate the impact of impaired upper limb motor function on trunk control and mobility in chronic post-stroke survivors so that the upper limb function should be enhanced as a core component of rehabilitation.

Methods: This analytical cross-sectional study was conducted from March 2021 to August 2021 at the Rehabilitation Center for the Physically Disabled in Peshawar, Pakistan. A total of 166 chronic stroke patients in the age range of 35 to 70 years and greater than 6 months post-stroke were included. To assess the impaired upper extremity, the Fugl-Meyer assessment upper extremity scale was used, and the trunk control and mobility were checked through functional reach test and timed up and go test, respectively.

Results: Among 166 stroke survivors having upper extremity impairment, 109(65.7%) participants' upper extremity was severely affected. Mobility levels measured through the timed up and go test were categorized and scored; accordingly, the participants showed a low risk of 22(13.3%), moderate 32(19.3%), high 72(43.4%), and very high risk of falls of 40(24%). Among 166 patients, 147(88.6%) had no trunk control which resulted in 8 times greater fall risk in 65(39.2%) participants. Thus, upper limb impairment is significantly associated with impaired trunk control and mobility (P=0.001).

Discussion: This study found about one-third of stroke survivors have a risk of falling, twothirds have upper extremity impairment, and two-thirds have no trunk control. This study showed a significant reduction in mobility level, balance, and posture balance in stroke survivors having upper extremity impairment. These findings suggest that stroke survivors with upper extremity impairment may use therapeutic intervention to minimize falls and enhance postural balance and trunk control.

.....

* Corresponding Author: Uzair Ahmad, Assistant Professor. Address: Department of Physiotherapy, College of Physical Therapy, Northwest Institute of Health Sciences, Peshawar, Pakistan. Tel: +(91) 915231605 E-mail: azmatuzair125@gmail.com

Highlights

• There is a significant association between upper extremity impairment and trunk control in post-stroke patients.

• A strong significant association was found between upper extremity impairment and the risk of falls in post-stroke patients.

• The upper extremity impairment and level of mobility were significantly associated with post-stroke patients.

Plain Language Summary

Stroke survivors face many complications, such as motor impairment, trunk control, abnormal sensation, and balance loss which causes limitations in their daily life activities. Motor impairment limits their walking ability, and social participation, thereby decreasing their chances of returning to professional work. After-stroke trunk control is an important early predictor of daily life activities. In addition, about one-fifth of stroke survivors have a marked decrease in mobility. This study analyzed 166 patients to find the impact of impaired upper limb motor function on trunk control and mobility in chronic post-stroke patients so that the upper limb function should be enhanced as a core component of rehabilitation. The result of this study showed that upper limb impairment in stroke patients is significantly associated with impaired trunk control, risk of falls, and mobility. Furthermore, these findings suggest that stroke survivors with upper extremity impairment may use therapeutic intervention to minimize falls and enhance postural balance and trunk control.

Introduction

troke is a major cause of illness globally. Each year, around 800000 patients suffer from strokes. Paralysis resulting from a stroke causes disability in stroke patients and most patients have difficulty walking [1]. Over the last 3 decades, the burden of stroke is

increasing in adults in the age range of 20 to 64 years [2]. Most people who had a stroke face many complications, such as the loss of motor power in muscles along with the sensation and balance loss which causes limitations in their daily life activities [3]. Immediately after the stroke, patients have weakness in their arm which does not resolve when spasticity sets after a few weeks or months, thereby further increasing the limitations with time and making treatment difficult [3]. Arm and hand movements are used in most common daily tasks [4]. A study of functional magnetic resonance imaging (fMRI) of subjects who recovered from hemiparetic stroke found that hemiparesis of the contralateral upper limb is the most common deficit after stroke, as it is experienced acutely in more than 80% of patients having stroke conditions. Meanwhile, more than 40% of stroke patients experience this disorder chronically [5]. Trunk control is the ability to remain upright during functional activities [6] Common impairment in postural balance occurs after a stroke and this postural impairment results in a fall [7]. After a stroke, trunk control is an important early predictor of daily life activities [8]. One study analyzed the swinging of arms during walking and according to the data, they concluded that during walking, gross movements are present in the shoulder, i.e. arm swing during gait occurs because of active muscle action and it is not a passive activity [9]. Another study conducted by Ortega et al. on the contribution of arms to the stability of gait concluded that arm swing contributes to lateral stabilization [10, 11]. The upper limb also helps in the recovery of balance after perturbation [12]. Also, it was observed in a study that individuals who are at risk of falls normally adopt a guarded posture after perturbation [13, 14]. During balance disturbance, individuals suddenly expand their arms to improve the stability of posture [15, 16]. Consequently, the developing literature on the impact of arms on postural balance, limiting arm movements will reduce the performance in mobility dynamic control of posture and during standing on a balance board the balance recovery is also impaired [17-20]. All outcomes show that the upper limbs have an important and significant influence on dynamic balance [21]. According to Patel et al., during static tandem standing, the mechanism of reducing postural sway is impaired when upper limb movements were restricted [21]. Due to a lack of past research, a further understanding can be obtained by measuring the impact of restricting arm movements on daily balance task performance which is commonly recommended with increasing task difficulty. Consequently, a recent research aimed to find out the effect of limiting or allowing movements of arms on postural sway during static standing tasks of different levels of difficulty [22].

According to an evidence, a number of studies showed a large number of studies showed the impact of lower limb function on postural balance, but only very limited evidence has shown the impact of upper limb motor function on postural balance [15]. This study aims to identify the impact of upper limb motor function impairment on trunk control and mobility in chronic post-stroke patients so that the upper limb function should be enhanced as a core component of rehabilitation.

Materials and Methods

A cross-sectional study was conducted from March 2021 to August 2021 in the Rehabilitation Center for the Physically Disabled in Peshawar, Pakistan. A total of 166 samples were calculated via the Rao Soft (online sample size calculator) with a 5% margin of error, a confidence level of 95%, a population size of 289, and a response distribution of 50%. The subjects were recruited through the convenience sampling technique. The inclusion criteria were as follows: Both genders in the age range of 35 to 70 years; being a post-stroke survivor; having chronic post-stroke survivor (after 24 weeks or 6 months); being of both known cases of ischemic and hemorrhagic stroke; being able to walk independently (without an assistive device); and scoring 21 or more on the Fugl-Meyer assessment-LE scale. The exclusion criteria were being a patient with a history of total knee replacement and total hip replacement surgery in the past 1 year; being a patient with diagnosed neurological disease other than strokes, such as the cognitive or perceptual deficit, multiple strokes, cerebellar lesions, visual field loss, diplopia, vertigo, dizziness; and having any acute medical illness and patients with amputations.

Upon the approval of the advanced studies & research board along with the ethical review board of Khyber Medical University Peshawar, Pakistan, the data was collected from rehabilitation center for the physically disabled. before data collection, permission was taken from the chief executive officer of the rehabilitation center for the physically disabled. An information document was provided to the stroke survivors who fulfilled the eligibility criteria and gave consent to participate in the study. To assess the impaired upper extremity (UE) and lower extremity (LE) motor function, the Fugl-Meyer Assessment (FMA) scale was used. The Fugl-Meyer Assessment-UE scale (FMA-UE) comprises 33 items, according to which upper limb motor function was assessed, the while Fugl-Meyer assessment-LE scale consists of 17 items, and its validity and reliability is shown by a study which demonstrates the sensitivity of 77% and specificity of 89% [23]. Patients scoring less than 28 on the FMA-UE scale indicated severe upper limb impairment. A study shows the reliability of this scale by interclass correlation coefficient at 0.95 and high concurrent validity with r=0.94-0.95 [23, 24]. For the trunk control assessment, a functional reach test was performed and mobility was checked through the time up and go test and the questionnaire was filled accordingly. All the evaluation tests were carried out by an experienced physical therapist who specialized in neurological physical therapy. All the subjects were assessed in the physical therapy outpatient department. After data collection, the data were analyzed via the SPSS software, version 25. The frequency and percentage were calculated for categorical data. The chi-square test was applied to find out the association among categorical variables.

Results

A total of 166 patients with a mean age of 53.0 ± 8.3 years participated in this study, out of which the reporting ischemic stroke prevalence was more than hemorrhagic, i.e. 71.7% vs 28.3%. The maximum number of stroke patients who had severe UE impairment was equal to n=109 (65.7%). A total of 166 stroke participants having UE impairment are reported in different categories of risk fall, that is, low risk of fall n=18 (10.8%), and individuals having 8 times greater fall risk were n=65 (39.2%). The highest number of stroke patients was n=147 (88.6%) with no trunk control. The patients were checked for mobility level on the bases of the time-up and go test, the maximum number of patients reported a high risk of falls at n=72 (43.4%) (Table 1).

The chi-square test was performed to assess the association between UE impairment and trunk control. There was a significant association between UE impairment and trunk control (P=0.001) (Table 2). Similarly, the association between UE impairment and the risk of falls was assessed by the chi-square test. A significant association was observed between UE impairment and the risk of falls (P=0.001) (Table 2). As mentioned in (Table 3), there was a significant association between UE impairment and level of mobility (P=0.001).

Discussion

The objectives of the present study were to investigate the effect of the impaired upper limb in post-stroke patients and its impact on trunk control and mobility. The present

Variables		Mean±SD/No. (%)	
Age (y)		53.0±8.3	
Turn of sharely		119(71.7)	
Type of stroke	Hemorrhagic	47(28.3)	
	Mild	30(18.0)	
Level of upper extremity impairment ^b	Moderate	27(16.3)	
	Severe	109(65.7)	
	Low risk	18(10.8)	
Risk of fall ^a	2 times greater risk	35(21)	
RISK OF TAIL"	4 times greater risk	35(21)	
	8 times greater risk	65(39.2)	
Trunk control ^b	No 147(88.6)	147(88.6)	
	Yes	19(11.4)	
	Very high risk of fall	40(24)	
Lougl of mobility (High risk of fall	72(43.4)	
Level of mobility ^c	Moderate risk of fall	32(19.3)	
	Low risk of fall	22(13.3)	

Table 1. Patients' characteristics (n=166)

^aMeasured on Fugl-Meyer assessment scale, ^bFunctional reach test, ^cTime up and go test.

ranian Rehabilitation Journal

study showed that individuals having UE impairment and risk of falls are in different categories, i.e. low risk of fall at 10.8% and greater risk of fall at 39.2%. Among all the participants, 88.6% had no trunk control and 11.4% had trunk control. Their level of mobility was checked through the time up and go test showed that 13% of the participants have a high mobility level and low risk of falls while 43% have a low mobility level and high risk of falls. Somewhat similar to the present study, a study from Northwestern University, Chicago, USA in 2018 showed a 28% prevalence of falls in subjects having upper limb loss and also contributing factors of frequent falls, which show

Table 2. Association of upper extremity impairment with trunk control and risk of fall

Variables		Level of	D 2		
		Severe Moderate I		Mild	— P ^a
Trunk control	No	105	20	22	0.001
	Yes	4	7	8	0.001
	Low risk	3	7	8	
Risk of fall	2 times greater risk	16	8	11	0.001
	4 times greater risk	38	6	4	0.001
	8 times greater risk	52	7	7	

^aChi-square test and significant level at P<0.05.

Iranian Rehabilitation Journal

Level of Upper Extremity Impairment	Level of Mobility on Bases of Time Up and Go Test/Risk of Fall				_ Total
	Very High	High	Moderate	Low	_ 10tai
Severe	35	52	19	3	109
Moderate	2	10	9	6	27
Mild	3	10	4	13	30
Total	40	72	32	22	166
P^{a}			0.001		

Table 3. Association upper extremity impairment and level of mobility

^aChi-square test and the significant level at P<0.05.

that upper limb loss proximally at the wrist have a greater number of falls. Some of them fell once while others fell twice. The study showed that about 70% of falls occur in a person with upper limb loss, cause of falls may be slips, trips, loss of balance, occur outside the environment, while not using stairs [25]. A Dutch study showed that in the injury surveillance system, where researcher records injuries treated at the emergency department along with in a follow-up study conducted in the year between 2003 and 2007, the frequency of single fall is most commonly found in older adults (about 33%) [26] and among stroke survivors, the ratio is about 40% [27, 28].

The present study found associations between UE impairment, trunk control, risk of falls, and 65% of the population have severe UE impairment, 63% have no trunk control, 39.2% have 8 times greater risk of falls, while patients having low mobility level have 43% high risk of fall (P=0.001) which shows a strong association between them. In line with our study, research from the University of Gothenburg, Sweden, used a cross-sectional study design using data from chronic stroke patients [29]. The data from the participants were taken from two data sources, namely a study by Carvalho et al. [30] and the Gothenburg very early supported discharge (GOTVED) [31]. The study shows that the UE affected post-stroke individuals and strongly correlates with postural control. Meanwhile, the Berg balance scale and time up and go test were used and showed a strong association between UE impairment and loss of postural balance. Berg balance scale showed that 21% have impaired balance while the time up and go test showed that 27% have impaired balance. Among all the participants, 5 patients were unable to participate in the time up and go test and were categorized as impaired balance. The patients who reported problems of postural balance were older than other patients who do not report any postural impairment. The study also reported better function of the UE measured

ver test and showed that it strongly cor-

Iranian Rehabilitation Journal

by the Fugl Mayer test and showed that it strongly correlated with a good balance of posture [32].

The current study also demonstrated decreased mobility level and speed of walk checked through time up and go test, where patients having severe upper limb impairment were having more low mobility level and increased fall risk as compared to individuals having mild upper limb impairment. Hong et al. (2020) reported a decrease in gait speed and step length when elbows are immobilized at 30, 90, or 120 degrees, and were having fixed orthosis while walking [33]. Moreover, in India, a study conducted in 2014 also exhibited a good association between upper and lower limbs' motor function level and balance. Meanwhile, the Berg balance scale items showed a good correlation with the components of the Fugl Meyer assessment UE scale [34]. This study is also in line with our research but shows a moderate correlation between impaired upper extremities and balance, the trunk control was not assessed, and the sample size was also very small (n=44 patients).

Our study showed that upper limb impairment can also be a risk factor in chronic post-stroke patients and can affect their mobility level. Among the studies that are in line with our research, a prospective cohort study was carried out and the mobility level was assessed by the river mead mobility index in the first and third year in the post-stroke period, and the decline was determined by >2 points on resonance magnetic image. Results show a decline in the status of mobility in about 21% of the population [35]. Most patients with UE impairment are unable to perform their daily life work normally and cannot participate in social activities because they do not achieve full recovery of their paralyzed arm [9]. This study also showed that patients were having severe upper limb impairment and decreased motor function assessed through the Fugl-Meyer assessment UE scale and it affected their daily life work.

Our study also showed a positive association between upper limb impairment and trunk control i.e. out of 109 patients having severe upper limb impairment, 105 patients' trunk control was also affected and their risk of falls was also high. A study in 2010 was conducted on 26 hemiplegic patients (11 men, 15 women) with shoulder subluxation, their balance was checked with and without wearing an arm sling. The results showed that the functional reach test was performed better with an arm sling which reduced the possible arm swing [36]. A study in 2014 states that the relationship between UE spasticity and balance presented that upper limb spasticity results in decreased balance confidence which contributes to an increased fear of falling [37]. This study also showed that patients having severe UE impairment have greater fall risk as compared to individuals having moderate or mild UE impairment.

Study limitations

In the present study, we gathered very limited data, because of the COVID-19 pandemic. Access to patients was difficult. This was a single-centered study, thus the sample size was small. Also, the pre-stroke mobility level and health status of patients may have affected their mobility and trunk control.

Recommendations

It is further recommended to carry out such studies on large populations in well-established rehabilitation centers and randomized controlled trials or other interventional studies should be carried out to find out the effect of upper limb impairment on balance and lower limb improvement. Proper attention should also be given to the early improvement of upper limb motor control to minimize falls and enhance balance.

Conclusion

This study found that about one-third of stroke, survivors have a risk of falling, two-thirds have UE impairment, and two-thirds have no trunk control. This study also demonstrated a significant reduction in mobility level, balance, and posture balance in stroke survivors having UE impairment. Thereover, it should be of clinical importance that LE impairment affects balance and posture, and the UE is affected by it. These findings suggest that a person with UE impairment may use therapeutic intervention to maximize postural balance. Furthermore, these findings suggest that a person have UE impairment may get benefits from regular monitoring and therapy by clinicians and go through targeted physical therapy to minimize falls and enhance postural balance and trunk control.

Ethical Considerations

Compliance with ethical guidelines

This study was approved by the Advanced Studies and Research Board (ASRB) and the Ethics Review Board of Khyber Medical University Peshawar (Code: KMU(IPM&R)/MSPT/2020/33258) and informed written consent has been obtained from individuals.

Funding

This research did not receive any grant from funding agencies in the public, commercial, or non-profit sectors.

Authors' contributions

Conceptualization and methodology: Erum Sattar; Investigation: Muheebur Rehman and Muhammad Adnan; Writing-original draft: Erum Sattar and Irum Shahid; Writing-review & editing: Uzair Ahmad and Subhan ur Rehman; Resources: Abdullah.

Conflict of interest

The authors declared no conflict of interest.

References

- Hwang YI, Yoon J. Changes in gait kinematics and muscle activity in stroke patients wearing various arm slings. Journal of Exercise Rehabilitation. 2017; 13(2):194-99. [DOI:10.12965/ jer.1734898.449] [PMID] [PMCID]
- [2] Krishnamurthi RV, Moran AE, Feigin VL, Barker-Collo S, Norrving B, Mensah GA, et al. Stroke prevalence, mortality and disability-adjusted life years in adults aged 20-64 Years in 1990-2013: Data from the global burden of disease 2013 Study. Neuroepidemiology. 2015; 45(3):190-202. [DOI:10.1159/000441098] [PMID]
- [3] Raghavan P. Upper limb motor impairment after stroke. Physical Medicine and Rehabilitation Clinics of North America. 2015; 26(4):599-610. [DOI:10.1016/j.pmr.2015.06.008] [PMID] [PMCID]
- [4] Shumway-Cook A, Woollacott MH. Motor control: Translating research into clinical practice. Philadelphia: Lippincott Williams & Wilkins; 2007. [Link]
- [5] Cramer SC, Nelles G, Benson RR, Kaplan JD, Parker RA, Kwong KK, et al. A functional MRI study of subjects recovered from hemiparetic stroke. Stroke. 1997; 28(12):2518-27. [DOI:10.1161/01.STR.28.12.2518] [PMID]

June 2023, Volume 21, Number 2

- [6] Wee SK, Hughes AM, Warner MB, Brown S, Cranny A, Mazomenos EB, et al. Effect of trunk support on upper extremity function in people with chronic stroke and people who are healthy. Physical Therapy. 2015; 95(8):1163-71. [DOI:10.2522/ ptj.20140487] [PMID]
- [7] Lord SR, Menz HB, Sherrington C. Home environment risk factors for falls in older people and the efficacy of home modifications. Age and Ageing. 2006; 35(Suppl 2):ii55-9. [DOI:10.1093/ageing/afl088] [PMID]
- [8] Verheyden G, Vereeck L, Truijen S, Troch M, Herregodts I, Lafosse C, et al. Trunk performance after stroke and the relationship with balance, gait and functional ability. Clinical Rehabilitation. 2006; 20(5):451-8.[DOI:10.1191/0269215505cr9550a] [PMID]
- [9] Elftman H. The function of the arms in walking. Human biology. 1939; 11(4):529. [Link]
- [10] Ortega JD, Fehlman LA, Farley CT. Effects of aging and arm swing on the metabolic cost of stability in human walking. Journal of Biomechanics. 2008; 41(16):3303-8. [DOI:10.1016/j. jbiomech.2008.06.039] [PMID] [PMCID]
- [11] Verheyden G, Nieuwboer A, Feys H, Thijs V, Vaes K, De Weerdt W. Discriminant ability of the trunk impairment scale: A comparison between stroke patients and healthy individuals. Disability and Rehabilitation. 2005; 27(17):1023-8. [DOI:10.1080/09638280500052872] [PMID]
- [12] Hof AL. The equations of motion for a standing human reveal three mechanisms for balance. Journal of Biomechanics. 2007; 40(2):451-7. [DOI:10.1016/j.jbiomech.2005.12.016] [PMID]
- [13] Lewek MD, Poole R, Johnson J, Halawa O, Huang X. Arm swing magnitude and asymmetry during gait in the early stages of Parkinson's disease. Gait & Posture. 2010; 31(2):256-60. [DOI:10.1016/j.gaitpost.2009.10.013] [PMID] [PMCID]
- [14] Meyns P, Bruijn SM, Duysens J. The how and why of arm swing during human walking. Gait & Posture. 2013; 38(4):555-62. [DOI:10.1016/j.gaitpost.2013.02.006] [PMID]
- [15] Meyns P, Desloovere K, Van Gestel L, Massaad F, Smits-Engelsman B, Duysens J. Altered arm posture in children with cerebral palsy is related to instability during walking. European Journal of Paediatric Neurology . 2012; 16(5):528-35 [DOI:10.1016/j.ejpn.2012.01.011] [PMID]
- [16] Pijnappels M, Kingma I, Wezenberg D, Reurink G, van Dieën JH. Armed against falls: The contribution of arm movements to balance recovery after tripping. Experimental Brain Research. 2010; 201(4):689-99. [DOI:10.1007/s00221-009-2088-7] [PMID]
- [17] Milosevic M, McConville KM, Masani K. Arm movement improves performance in clinical balance and mobility tests. Gait & Posture. 2011; 33(3):507-9. [DOI:10.1016/j.gaitpost.2010.12.005] [PMID]
- [18] Hébert-Losier K. Clinical implications of hand position and lower limb length measurement method on y-balance test scores and interpretations. Journal of Athletic Training. 2017; 52(10):910-7. [DOI:10.4085/1062-6050-52.8.02] [PMID] [PMCID]
- [19] Hill MW, Wdowski MM, Pennell A, Stodden DF, Duncan MJ. Dynamic postural control in children: Do the arms lend the legs a helping hand? Frontiers in Physiology. 2019; 9:1932. [DOI:10.3389/fphys.2018.01932] [PMID] [PMCID]

- [20] Shafeie M, Manifar S, Milosevic M, McConville KM. Arm movement effect on balance. Annual International Conference of the IEEE Engineering in Medicine and Biology Society. 2012; 2012:4549-52. [DOI:10.1109/EMBC.2012.6346979] [PMID]
- [21] Patel M, Buckwell D, Hawken M, Bronstein AM. Does outstretching the arms improve postural stability? Neuroscience Letters. 2014; 579:97-100. [DOI:10.1016/j.neulet.2014.07.010] [PMID]
- [22] Boström KJ, Dirksen T, Zentgraf K, Wagner H. The contribution of upper body movements to dynamic balance regulation during challenged locomotion. Frontiers in Human Neuroscience. 2018; 12:8. [DOI:10.3389/fnhum.2018.00008] [PMID] [PMCID]
- [23] Lundquist CB, Maribo T. The fugl-meyer assessment of the upper extremity: Reliability, responsiveness and validity of the Danish version. Disability and Rehabilitation. 2017; 39(9):934-9. [DOI:10.3109/09638288.2016.1163422] [PMID]
- [24] Zeltzer L. Fugl-meyer assessment of sensorimotor recovery after stroke (FMA) [Internet]. 2010 [Updated 2023 June 28]. Available from: [Link]
- [25] Major MJ. Fall prevalence and contributors to the likelihood of falling in persons with upper limb loss. Physical Therapy. 2019; 99(4):377-87. [DOI:10.1093/ptj/pzy156] [PMID] [PMCID]
- [26] Hartholt KA, van Beeck EF, Polinder S, van der Velde N, van Lieshout EM, Panneman MJ, et al. Societal consequences of falls in the older population: Injuries, healthcare costs, and long-term reduced quality of life. The Journal of Trauma. 2011; 71(3):748-53. [DOI:10.1097/TA.0b013e3181f6f5e5] [PMID]
- [27] Kerse N, Parag V, Feigin VL, McNaughton H, Hackett ML, Bennett DA, et al. Falls after stroke: Results from the Auckland regional community stroke (ARCOS) study, 2002 to 2003. Stroke. 2008; 39(6):1890-3. [DOI:10.1161/STROKEA-HA.107.509885] [PMID]
- [28] Belgen B, Beninato M, Sullivan PE, Narielwalla K. The association of balance capacity and falls self-efficacy with history of falling in community-dwelling people with chronic stroke. Archives of Physical Medicine and Rehabilitation. 2006; 87(4):554-61. [DOI:10.1016/j.apmr.2005.12.027] [PMID]
- [29] Bernhardt J, Hayward KS, Kwakkel G, Ward NS, Wolf SL, Borschmann K, et al. Agreed definitions and a shared vision for new standards in stroke recovery research: The stroke recovery and rehabilitation roundtable taskforce. International Journal of Stroke. 2017; 12(5):444-50. [DOI:10.1177/1747493017711816] [PMID]
- [30] Carvalho C, Sunnerhagen KS, Willén C. Walking speed and distance in different environments of subjects in the later stage post-stroke. Physiotherapy Theory and Practice. 2010; 26(8):519-27.[DOI:10.3109/09593980903585042] [PMID]
- [31] Sunnerhagen KS, Danielsson A, Rafsten L, Björkdahl A, Axelsson ÅB, Nordin Å, et al. Gothenburg very early supported discharge study (GOTVED) NCT01622205: A block randomized trial with superiority design of very early supported discharge for patients with stroke. BMC Neurology. 2013; 13:66.[DOI:10.1186/1471-2377-13-66] [PMID] [PMCID]
- [32] Rafsten L, Meirelles C, Danielsson A, Sunnerhagen KS. Impaired motor function in the affected arm predicts impaired postural balance after stroke: A cross sectional study. Frontiers in Neurology. 2019; 10:912. [DOI:10.3389/fneur.2019.00912] [PMID] [PMCID]

- [33] Hong SH, Jung SY, Oh HK, Lee SH, Woo YK. Effects of the immobilization of the upper extremities on spatiotemporal gait parameters during walking in stroke patients: A preliminary study. BioMed Research Internationa. 2020; 2020:6157231. [DOI:10.1155/2020/6157231] [PMID] [PMCID]
- [34] Arya KN, Pandian S, Abhilasha CR, Verma A. Does the motor level of the paretic extremities affect balance in poststroke subjects? Rehabilitation Research and Practice. 2014; 2014:767859. [DOI:10.1155/2014/767859] [PMID] [PMCID]
- [35] van de Port IG, Kwakkel G, van Wijk I, Lindeman E. Susceptibility to deterioration of mobility long-term after stroke: A prospective cohort study. Stroke. 2006; 37(1):167-71. [DOI:10.1161/01.STR.0000195180.69904.f2] [PMID]
- [36] Acar M, Karatas GK. The effect of arm sling on balance in patients with hemiplegia. Gait & Posture. 2010; 32(4):641-4. [DOI:10.1016/j.gaitpost.2010.09.008] [PMID]
- [37] Phadke CP, Moody T, Mochizuki G, Gage W, Ismail F, Boulias C. Relationship between spasticity and balance confidence in persons post-stroke. Archives of Physical Medicine and Rehabilitation. 2014; 95(10):e15. [DOI:10.1016/j. apmr.2014.07.024]